

## Format for ANSWERING REVIEWERS

March 29, 2016



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 25531review.doc).

**Title:** Clinical study of dezocine combined with propofol in the anesthetization of patients undergoing indolent colonoscopy for gastrointestinal function

**Author:** Bin-bin Xu, Xiao-linag Zhao, Gui-ping Xu, Department of Anesthesiology People's Hospital of Xinjiang Autonomous Region Xinjiang urumqi city in China

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 25531

The manuscript has been improved according to the suggestions of reviewers:

1 The data has been check again.

2 Revision has been made according to the suggestions of the reviewer

The manuscript is very well written, but some editings needed. Some language errors should be corrected, and the list should be updating.

3 References are updated, and well discussed.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Bin-bin Xu

A handwritten signature in black ink, appearing to be '徐斌斌' (Xu Binbin).

Department of Anesthesiology

People's Hospital of Xinjiang Autonomous Region

Xinjiang urumqi city in China

E-mail: 56089778@qq.com